Author's response to reviews

Title: The association between alcohol use, alcohol use disorders and tuberculosis (TB). A systematic review.

Authors:

Jürgen Rehm (jtrehm@aol.com)
Andriy V Samokhvalow (avsamokhvalov@yahoo.ca)
Manuela G Neuman (manuela.neuman@utoronto.ca)
Robin Room (RobinR@TURNINGPOINT.ORG.AU)
Charles Parry (charles.parry@mrc.ac.za)
Knut Lönnroth (lonnrothk@who.int)
Jayadeep Patra (jaydeep.patra@gmail.com)
Vladimir Poznyak (poznyakv@who.int)
Svetlana Popova (lana_popova@camh.net)

Version: 4 Date: 17 October 2009

Author's response to reviews: see over
Dear editors,

please find attached the revised manuscript. We believe we were able to respond to the comments of the reviewers and the Editorial Board Member. Part of problem was that in our paper, we had done two systematic reviews, as described in the methods. We hope that the current version actually clarified this point.

All 9 authors contributed to the revision of the article, and all of them approved of the final version. Getting all the revision done in this situation with authors in three continents takes a bit more time, which was the reason we had requested more time.

We look forward to your evaluation of the manuscript.

Best wishes
Jürgen Rehm, Ph.D.
Professor and Chair, Addiction Policy, Dalla Lana School of Public Health, University of Toronto, Canada
Senior Scientist and Co-Head, Section Public Health and Regulatory Policies, Centre for Addiction and Mental Health, Toronto, Ontario, Canada
Professor, Dept. of Psychiatry, Faculty of Medicine, University of Toronto, Canada
Head, Epidemiological Research Unit, Technische Universität Dresden, Klinische Psychologie & Psychotherapie

mailing address:
Centre for Addiction and Mental Health
33 Russell Street, Room 2035
Toronto, Ontario
Canada M5S 2S1
Tel: ++1 416 535 8501 ext. 6907
Fax: ++1 416 260 4156